

Article





Government Intervention and Automobile Industry Structure: Theory and Evidence from China

Lin Ma¹, Qinchuan Du² and Tian Wu^{3,4,5,*}

- ¹ School of Banking & Finance, University of International Business and Economics, Beijing 100029, China
- ² Academy of Macroeconomic Research, National Development and Reform Commission, Beijing 100038, China
- ³ NCMIS, Academy of Mathematics and Systems Science, Chinese Academy of Sciences, Beijing 100190, China
- ⁴ School of Economics and Management, University of Chinese Academy of Sciences, Beijing 100190, China ⁵ Key Laboratory of Big Data Mining and Knowledge Management, Chinese Academy of Sciences
- ⁵ Key Laboratory of Big Data Mining and Knowledge Management, Chinese Academy of Sciences, Beijing 100190, China
- * Correspondence: wutian@amss.ac.cn; Tel.: +86-10-8254-1644

Received: 25 July 2019; Accepted: 23 August 2019; Published: 29 August 2019



Abstract: The development of the automobile maintenance industry less developed to satisfy the increasing demand for automobile maintenance service as the automobile manufacturing industry increased rapidly in China. This is not conducive to the sustainable development of the automobile industry. Besides the factors of market behavior that can affect the automobile industry structure, like an investment, operation structure or economic development stage, the structure is also influenced by government intervention. We investigated the unbalanced development of automobile structure from the perspective of government incentives, and provide a logical framework for analyzing the industrial policies on the automobile industry. We first established a two-sector theoretical model with government intervention, and we found that the governments' GDP incentive induced the biased intervention policy. More preferential policies are given to enterprises of automobile manufacturing industries as they contribute more to intermediate goods and capital. The greater the government's GDP incentive, the more biased the intervention will be. Then we test the differential impact of GDP incentive on tax avoidance of the two kinds of firms empirically. The empirical results show that GDP incentive of the government induced more preferential treatment to automobile manufacturing enterprises, and thus, increased their tax avoidance. This phenomenon is more significant in SOEs, larger firms and firms belong to local governments. Understanding the incentive and implementation of industrial policy can help us know the evolution of automobile industrial structure better, and then improve industrial policy better to promote the transformation and upgrading of automobile industrial structure.

Keywords: automobile manufacturing; automobile maintenance; government incentive; government intervention; tax avoidance

1. Introduction

The economic development process of various countries in the world shows that vehicle ownership gradually increases with the development of economy, especially GDP per capita [1]. Although the automobile manufacturing industry increased rapidly in China, automobile maintenance industry less developed compared with developed countries and has not received enough attention [2]. The auto care industry is developing quickly in the United States in recent years. According to the data of Auto Care Association, the number of people employed in the auto care industry is 4.6 million, and the number of automobiles in operation is 278.6 million in early 2018, and the average annual sales value

of auto care industry is now 381 billion. Chain operation mode, which has been developed rapidly in 30 years, was adopted in automobile after-sales service companies in the United States [3]. The supply chain not only includes the automobile manufacturer, but also the after-sales service market of auto parts supply and an integrated maintenance service provider [4]. This pattern not only makes the price of service more transparent, but also can integrate the auto parts resources of each brand and break the vertical monopoly [5].

The situation of auto care is less encouraging in China. According to the China Automotive Aftermarket Blue Book (2013–2017) [6], the average annual growth rate of China's automotive aftermarket will exceed 30% in the future, and the market size is expected to exceed one trillion Yuan after 2018. According to the news of research and markets [7], Chinese automotive aftermarket revenue is expected to record a compound annual growth rate of 7.7%, increasing from \$290.44 billion in 2017 to \$523.80 billion in 2025. During this period, the vehicles in operation for Chinese passenger vehicles are expected to grow from 185 million units to 401.7 million units. However, the development of automotive aftermarket is not sufficient to satisfy the increasing demand for automobile maintenance service in China. 4S stores (store for automobile sale, spare parts, service, and survey) is the leading form of after-sales service market in China [8], and it needs a large investment to develop well. However, many of them with limited sales in the market are unable to cover their costs with intense competition and the gradual increase of investment [6]. The development of domestic automobile maintenance enterprises in the initial stage will also be impacted by the external competitive environment because foreign enterprises bring more new technologies and new business models when they enter the market.

The ignorance of maintenance and service will have a negative impact on the automobile industry, which is not conducive to the sustainable development of the automobile industry. Firstly, the quality and development of auto maintenance industry will affect customer loyalty, which has been paid attention by the American automobile industry [9,10]. There are a large number of single stores and brand chains providing high-quality maintenance and customer service in the United States, and automobile service is insurance of financial and strategic advantages to help the company survive in this highly competitive industry [11,12]. Secondly, from the perspective of market sustainability, automobile maintenance will satisfy and promote the development of new kinds of vehicles. The market share of shared vehicles, electric vehicles and autonomous driving is increasing, and maintenance process of these kinds of vehicles is different with the conventional ones, for example, the annual maintenance cost of vehicles will increase as the annual mileage of shared cars increases. In addition, customers' expectations and requirements on the automotive maintenance industry are also constantly rising with the development of economy, science and technology. So it is necessary for the automotive service industry to constantly improve its development mode and introduce new technologies, which presents challenges and opportunities for the whole industry. Thirdly, lack of attention to maintenance will make China's maintenance industry lag behind in the Internet era. Nowadays, the popularization of the internet brought business opportunities for shopping and O2O (Online to Offline) service industry [13,14], and the automobile maintenance industry should also grasp the opportunities of the internet era and realize the transition of traditional industries. Finally, from the perspective of sustainable regional growth, the diversity of aftermarket growth in different regions reflects the market maturity to a large extent. The growth rate of emerging markets will exceed that of mature markets, and the rapid growth of China's vehicle ownership will drive the share of Asia in the global aftermarket. According to the experience of North American and European markets, the more developed the post-market is, the higher the degree of integration in the transaction level of the industry is. Therefore, it is of great significance to understand the reasons of the slow development of the automobile maintenance industry in China and to make the automobile maintenance and automobile manufacturing industry develop and integrate in a balanced way. What's more, automobile maintenance industry is an important part of the service industry, which is critical to national competitiveness and industrial structure [15,16]. The appropriate industrial structure adjustment will promote economic growth because of the "structure bonus" brought by the movement of production

factors in different sectors [17,18], especially for developing countries [19,20]. Although it's right to develop the automobile manufacturing industry more than the automobile maintenance industry in the initial phase of the development of the automotive industry, China should pay more attention to the balanced development of the two industries after passing this first phase.

As an important part of the service industry, there are some reasons why China's automobile maintenance industry has not developed enough. Firstly, the imperfect investment and financing environment lead to the fact that most maintenance firms lack external financial support and the investment is far away from enough for development. Secondly, China is still in early stage of vehicle ownership development process, and the ratio of vehicle ownership per thousand people is only 172 according to the data of World Road Statistics dataset (WRS) [21]. This number is far away from saturation level, 807, according to Dargay and Sommer [22], and there is still a lot of space for the automobile industry to develop. Thirdly, some researchers believe customer recognition is the most serious barrier at present in the external environment [9,10]. Most of China's automobile maintenance businesses do not pay sufficient attention to information resources, and they have not yet formed a unique business model that is closely connected with the market [6]. What's more, some market factors that affect industrial structures, such as the change of demand, technology improvement, endowment and trade [23–26], can affect the automobile maintenance industry as well. According to the theory of development economics, industrial structure changes with economic development; however, the industrial structure varies in countries that share similar development stages. Although the economy grows rapidly, industry accounts for the largest share of GDP for years until 2015 in China, and the ratio of the service industry is still far away from that of developed countries, even some developing countries. According to the data of Statistical Yearbook of the World [27], service sector accounts 70.9% on average all over the world, 74.3% in the high-income countries in 2010, and 55.6% in middle income countries, 55.5% in low income countries, while, it is only 44.6% in China. So, are there any other reasons that can explain these facts?

Since the marketization degree in China is not very high, and the government plays a great role in the economy, we think that government intervention is another important factor that can affect industry structure in China. Some researchers proposed that in early stage of economic development, industrial policies may accelerate the industrial structural change and improve resource allocation [28,29]. It is also found that the industrial policy can enhance competition among enterprises and promote the growth of enterprises [30]. Local governments are important participants in promoting China's economic development in the process of China's industrial transformation. Therefore, in addition to the natural evolution of market behavior, the industrial structure is also subject to government intervention. We try to explain the problem from the perspective of government incentives and provide a logical framework for analyzing the industrial structure. Some researchers proposed incentives for self-interest and for electoral success in public choice theory [31,32]. Some researchers believe that local governments have fiscal revenues, GDP and promotion incentives under the centralized political and decentralized fiscal institute in China [33]. Thus, many stimulus measures are adopted to boost economies [34,35] and then affected industrial structure [36]. However, the mechanism of the process has not been systematically studied because existing researches pay more attention to the evaluation of policy effects. Therefore, we try to explain the mechanism of the impact of the government incentives on automobile industrial structure.

The government can use different kinds of interventions, which take many forms, including explicit subsidies and bonus, such as research and development subsidies [37–39], but more of them are invisible, such as granting loan guarantees and loan preferences to enterprises to reduce their capital costs and financing constraints [40]. Another example is effective tax rates, which can be affected by tax preference and taxation intensity so that a firm's tax burden can be affected as well, although the tax rate is legal [41,42]. This kind of intervention changes with government incentives and varies in different enterprises.

We found both theoretically and empirically that the governments' GDP incentive induced the biased intervention policy on automobile manufacturing and maintenance industries. In the theoretical model, the government's GDP incentive leads to the more patience of the government than the consumer. The government doesn't care about current consumption, but care about whether enough investment and capital can be generated to promote economic growth. Thus, the enterprises of different industries that have different contributions to investment and capital accumulation will inevitably be subject to different government intervention. More preferential subsidies are given to automobile manufacturing firm whose production serve as intermediate goods and capital. Moreover, the greater the government's GDP incentive, the more biased the intervention will be. Then we test the differential impacts of GDP incentive on tax avoidance of the two kinds of firms empirically. The empirical results show that the government GDP incentive induced more preferential treatment to automobile manufacturing enterprises, and thus, increased their tax evasion compared with automobile maintenance firms.

Our study makes several contributions to the existing literature. First, we explained why China's automobile manufacturing and maintenance industry is unbalanced from the perspective of government intervention and incentives. Some researchers analyzed government intervention instruments and their effects on economic variables, such as resource misallocation, total factor productivity and costs [43,44], but little literature involves the motivation of government policies. The literature on optimal policy [30-32], which adds government policy in the general equilibrium model are more about the fiscal policy, monetary policy or tax policy on consumption or capital; little is about industrial policy in different sectors. Therefore, we investigated the impact of government incentive on industrial policy and industry structure, and provide a framework to study the interaction of government and the market, which also applies to other countries. Second, the growth rate of GDP is converging from the perspective of the development economics [45], and since automobile industry has always been a pillar industry and in China, how to promote the automobile industry to develop in a balanced way is crucial to the sustainable development of the automobile industry and even the whole economy in the future. Although some researchers focus on sustainable district development [46], balanced development of industry structure is also important. Finally, the research about China's automobile industry mainly focus on the manufacturing industry currently, such as the prediction of vehicle ownership [1,47,48], study on life cycle cost of electric vehicle [49–52], the bottleneck in the development of electric vehicles [53–55], and research on the development and market scale of ride-hailing [56,57]. There is little literature on automobile maintenance industry; however, we need to fully understand the development of the automobile industry, including the maintenance industry to promote sustainable development.

The remainder of the paper is organized as follows. Section 2 presents the theoretical model and the development of the empirical hypothesis. Section 3 introduces the empirical model, variable definitions and data resources. Section 4 provides the empirical evidence. Section 5 provides several robustness checks to make sure our results are robust. Section 5 concludes and presents policy implications.

2. Theoretical Model and Empirical Hypothesis

We build a general equilibrium model with government incentive to investigate the effect of government intervention on the automobile industry. The general equilibrium model is widely used in modern economic research, especially the macroeconomic research about economic growth and industry structure [15,17,18,23–25]. Furthermore, the method of adding government policy in the general equilibrium model is usually used in the literature of optimal policy [30–32]. However, the literature is more about the fiscal policy, monetary policy or tax policy on consumption or capital, little is about industrial policy in different sectors. Thus, we also investigate the impact of government intervention on the automobile industry structure and its mechanism in this framework.

2.1. Theoretical Model

We assume that there are four participants in the economy: A representative consumer, two representative producers (i.e., one is an automobile manufacturing firm and the other is automobile maintenance firm) and a government. The market is completely competitive. The products of automobile manufacturing firm produce the finished automobile and auto parts, which can be formed capital in the economy, while the products or service of automobile maintenance firm can be consumed. Thus, the automobile manufacturing firm is more likely to be the upstream firm, and while the automobile maintenance firm is a downstream firm in the input-output production connections. Both of the two firms use capital and labor as input in the production process. We use subsidies as tools of government intervention and omit other intervention methods, such as tax rate. The subsidy rate for the automobile manufacture firm is denoted as s_1 , and the subsidy rate for the automobile manufacture firm is denoted as s_2 . The subsidy rates can be explained as the degree of government preferential, such as the relaxation of taxation [42]. The representative consumer is the providers of capital and labor. He provides capital and labor to producers and obtains rents and wage at the same time. Then the consumer uses the income to consume maintenance service to maximize utility for all periods of time. The utility maximization problem of the representative consumer can be expressed as:

$$\max_{c_t, K_{t+1}} u = \sum_{t=0}^{\infty} \beta^t u(c_t), \tag{1}$$

$$c_t + p_t(K_{t+1} - K_t + \delta K_t) \le r_t K_t + w_t L - T_t,$$
 (2)

here, u is the utility of the consumer, and $u(\cdot)$ is the utility function; c_t is the consumption of the consumer at time t; K_t is the capital stock at time t; L is the fixed labor supply per period; β is the discount rate, and δ is the depreciation rate of capital. Since there are only two products in the economy, we might as well set the price of the automobile maintenance service as one and the price of the automobile manufacturing production as p_t . The wages of the labor and the rent of capital are w_t and r_t , respectively. T_t is the lump-sum tax on consumer's income. We list the definition of each parameter in Table 1. If we set the Lagrangian multiplier of each period is λ_t , then the first-order condition of the consumer problem is:

$$u'(c_t) = \lambda_t,\tag{3}$$

$$\lambda_t p_t = \beta \lambda_{t+1} (p_{t+1} + r_{t+1} - \delta p_{t+1}).$$
(4)

Thus, the Euler Equation is obtained from Equations (3) and (4):

$$\frac{p_t}{r_{t+1} + (1-\delta)p_{t+1}} = \frac{\beta u'(c_{t+1})}{u'(c_t)}.$$
(5)

We assume that the production function of the automobile maintenance firm is $F_1(K_{1t}, L_{1t})$ and the production function of the automobile manufacturing firm is $F_2(K_{2t}, L_{2t})$. The production functions satisfy $F_{iK} > 0$, $F_{iL} > 0$, $F_{iLL} < 0$, $F_{iKL} > 0$, i = 1, 2. The producers in each sector maximize the profits of each period:

$$\max_{K_{1t},L_{1t}} \pi_1 = (1 + s_{1t})F_1(K_{1t}, L_{1t}) - w_t L_{1t} - r_t K_{1t},\tag{6}$$

$$\max_{K_{2t},L_{2t}} \pi_2 = p_t (1 + s_{2t}) F_2(K_{2t}, L_{2t}) - w_t L_{2t} - r_t K_{2t}, \tag{7}$$

here, π_1 and π_2 represent the profit of automobile maintenance and manufacturing firms, respectively. Then the first-order conditions for the capital of the automobile maintenance firm and automobile manufacture firm are:

$$(1+s_{1t})F_{1K} = r_t, (8)$$

$$p_t(1+s_{2t})F_{2K} = r_t, (9)$$

here, we omit the subscript of time *t* of the production functions for briefness. The first-order conditions for labor of the automobile maintenance firm and automobile manufacture firm are:

$$(1+s_{1t})F_{1L} = w_t, (10)$$

$$p_t(1+s_{2t})F_{2L} = w_t. (11)$$

We can know from Equations (8)–(11) that the government can influence the marginal production of labor and capital and then the production decisions of the firms, and then determine the capital flows between the two sectors by setting different subsidy rates. The government's subsidy expenditure comes from the lump-sum tax T_t imposed on consumers. Thus, the government's budget constraint is

$$T_t = s_{1t}F_1(K_{1t}, L_{1t}) + s_{2t}p_{2t}F_2(K_{2t}, L_{2t}).$$
(12)

As a social planner, the government's goal is certainly to maximize social welfare. However, the local governments have the promotion pressure in their career under the decentralization institute in China, so their goal is not entirely social welfare maximization, but the short-term GDP in their tenure. This incentive makes the government's time preference different from the consumer's in the market. The government is more willing to spend in the future and save currently to promote short-term growth of GDP. We set the government's short-term GDP incentive intensity to be *a*, which satisfies $1 < a < 1/\beta$. Then the discount rate of government is $a\beta$, and the larger the value of *a*, the greater the GDP incentive the government has. Therefore, the government's objective function is:

$$\max u_g = \sum_{t=1}^{\infty} (a\beta)^t u(c_t), \tag{13}$$

here, u_g is the utility of the government. Given the subsidy rates s_{1t} and s_{2t} for the automobile maintenance firm and automobile manufacturing firm, the equilibrium of market can be determined by Equations (2), (5), (8)–(11) and the market clear conditions of the products of the two sectors, capital and labor, which is the Equations (14)–(17) below. The market clear condition for automobile maintenance service market is:

$$F_1(K_{1t}, L_{1t}) = c_t. (14)$$

The market clear condition for automobile manufacturing market is:

$$F_2(K_{2t}, L_{2t}) = K_{1,t+1} + K_{2,t+1} - (1 - \delta)(K_{1t} + K_{2t}).$$
(15)

The market clear condition for capital is:

$$K_{1t} + K_{2t} = K_t. (16)$$

The market clear condition for labor is:

$$L_{1t} + L_{2t} = L. (17)$$

This system expressed the dynamic equilibrium relationships for variables { c_t , K_{1t} , K_{2t} , K_t , L_{1t} , L_{2t} , p_t , r_t , w_t , T_t }. Then the government will choose the optimal subsidy rates s_{1t} and s_{2t} constrained by the market equilibrium conditions. We solve this problem with the steps below. Firstly, using the Euler Equation, which is Equation (5) and budget constraint, which is Equation (2), we change the consumer's budget constraint, which is Equation (2), of each period into budget constraint of all periods (please see the detail calculation in Appendix A part), which is:

$$\sum_{t=0}^{\infty} \beta^t \, u'(c_t)(c_t - w_t L + T_t) = u'(c_0)[r_0 K_0 + p_0(1 - \delta)K_0]. \tag{18}$$

We denote the right hand of Equation (18) as $A = u'(c_0)[r_0K_0 + p_0(1 - \delta)K_0]$, which is constant. Then the government will optimize Equation (13) conditioning on Equations (14)–(18), and choose the optimal value of variables { c_t , K_{1t} , K_{2t} , L_{1t} , L_{2t} }. The Lagrange function of this problem is:

$$\Gamma = \sum_{t=0}^{\infty} (a\beta)^{t} \begin{cases} u(c_{t}) + \eta_{1t} [F_{1}(K_{1t}, L_{1t}) - c_{t}] \\ + \eta_{2t} [F_{2}(K_{2t}, L_{2t}) - K_{1t+1} - K_{2t+1} + (1-\delta)(K_{1t} + K_{2t})] \\ + \eta_{3t} [L - L_{1t} - L_{2t}] \end{cases}, \quad (19)$$

$$+ \sum_{t=0}^{\infty} [\beta^{t} \phi u'(c_{t})(c_{t} - w_{t}L + T_{t})] - \phi A$$

here, the parameters η_{1t} , η_{2t} , η_{3t} , ϕ are Lagrange multipliers for the constraints. We can get the dynamic equilibrium system for { c_t , K_{1t} , K_{2t} , L_{1t} , L_{2t} , η_{1t} , η_{2t} , η_{3t} , ϕ } using the first order conditions of variables c_t , $K_{1, t+1}$, $K_{2,t+1}$, L_{1t} , L_{2t} of Equation (19) and constraints, which is Equations (14), (15) and (17). Specially, according to the analysis of the first order conditions at steady state we can get this conclusion below.

Proposition 1. When a > 1, we have $s_2 > 0$; the greater the government's GDP incentives, the larger the subsidy s_2 to the automobile manufacturing sector, and the greater the gap of government's preferential support for automobile manufacturing sector and that for automobile maintenance sector.

Proof. We only focus on the first order conditions of variables $K_{1, t+1}$, $K_{2,t+1}$ and c_t , at the steady state, which means we can omit the subscript of time, and they can be expressed as:

$$\eta_2 = a\beta [\eta_1 F_{1K} + (1 - \delta)\eta_2], \tag{20}$$

$$\eta_2 = a\beta [\eta_2 F_{2K} + (1 - \delta)\eta_2], \tag{21}$$

$$\eta_1 = u'(c) + \frac{\phi u''(c)(c - wL) + u'(c)}{a^t}.$$
(22)

From the steady state of Euler Equation, which is Equation (5), we can get:

$$\frac{1}{\beta} = 1 - \delta + \frac{r}{p}.$$
(23)

From the first order condition of the automobile maintenance sector, which is Equation (8), at the steady state, we can get:

$$F_{1K} = r/(1+s_1). \tag{24}$$

From the first order condition of the automobile manufacturing sector, which is Equation (9), at the steady state, we can get:

$$pF_{2K} = r/(1+s_2). \tag{25}$$

Substituting the conditions of Equations (8) and (23) into Equation (20), we can get:

$$\frac{1}{a\beta} - \frac{1}{\beta} + \frac{r}{p} = \frac{\eta_1 r}{\eta_2 (1 + s_1)},$$
(26)

and substituting the conditions of Equations (9) and (23) into Equation (21), we can get:

$$\frac{1}{a\beta} - \frac{1}{\beta} + \frac{r}{p} = \frac{r}{p(1+s_2)}.$$
(27)

Then from Equation (27), since we assume as a > 1, the subsidy rate for the automobile manufacturing sector $s_2 > 0$, and s_2 increases with the increasing of a, which means the government

indeed give subsidy to the automobile manufacturing firm, and the subsidy increases with the increasing of short term GDP incentive.

Then we will analyze the subsidy to the automobile maintenance firms. Dividing Equation (27) by Equation (26), we can get:

$$\frac{1+s_1}{1+s_2} = \frac{p\eta_1}{\eta_2}.$$
(28)

We can know that η_1 decreases with parameter *a*, while η_2 does not change with parameter *a* according to Equation (22) assuming u''(c) is small enough, and the term $\phi u''(c)(c - wL) + u'(c)$ is positive. This means the shadow value of automobile maintenance service decrease when the government' GDP incentive is large. So $\frac{1+s_1}{1+s_2}$ will decrease with *a*, which means the gap of government's preferential support for automobile manufacturing sector and automobile maintenance sector increases with government's GDP incentive. \Box

These results can be interpreted as that the government pays more attention to future consumption and production, due to short-term GDP incentives and the limited tenure, which means the government utility is different with the consumer's because the consumer prefers the current consumption. Then the government subsidizes the firms in the automobile manufacturing sector that can be formed as capital and then more capital will produce more product and GDP because the capital is the necessity of the production process and has the multiplier effect in the process. Moreover, the larger the short-term GDP incentives, the larger the subsidies on the automobile manufacturing sector, and the larger gap between the subsidy given to the automobile manufacturing firm and the subsidy given to the automobile maintenance firm. That's why the government with short-term GDP incentive will intervene differently in the different sectors.

In the empirical part in Section 3, we will test the positive correlation of the government's short-term GDP incentives on the biased subsidy between the firms of automobile manufacture sector and firms of automobile maintenance sector. So we firstly propose the empirical hypothesis in Section 2.2 before the empirical research design.

2.2. Empirical Hypothesis

According to the theoretical analysis in Section 2.1, we know that when the government has short-term GDP incentives, he intervenes differently in automobile industries. In order to provide more capital for the entire economy to drive more output of all industries because of the multiplier effect of the capital in macroeconomics, the government will support the development of the automobile manufacturing sector with more subsidies, while the subsidies implemented in the automobile maintenance industry will be relatively less. The differential subsidies can reduce consumption and promote capital accumulation, thus, contribute to GDP growth. To prove the mechanism of our theoretical model and conclusion, we will give some empirical evidence about how the government's short term GDP incentive can affect the biased government intervention between the firms of automobile manufacture sector and firms of automobile maintenance sector. Furthermore, we will find the appropriate proxies of government intervention and government's short-term GDP incentives.

As for the measure of government intervention, the government subsidy is a kind of government intervention, and it is usually carried out in the form of preferential treatment or hidden subsidies for firms, which cannot be seen directly, however, the preferential treatment can be reflected in the degree of tax avoidance of firms. Although the tax collection is stipulated by the law of tax, the tax law also allows the government to implement tax reduction and tax preference accordingly, which induced that the actual tax rates of enterprises may not the actual tax rate ruled by the tax law. In addition, the government can use different taxation intensity; thus, the degrees of tax evasion of different enterprises are different. Thus, when the government has more GDP incentive, more preferential treatment will be given to automobile manufacturing enterprises, and the tax avoidance of these enterprises should be larger. So we use the tax avoidance of the firm to represent the

degree of government intervention, and the larger tax avoidance, the large degree of government intervention. Since the government intervention is biased to automobile manufacturing enterprises and the biased gap increases with the government's GDP incentives, we have the following empirical hypothesis below.

Hypothesis 1. *The higher government's GDP incentives will increase the gap of tax avoidance between the automobile manufacturing enterprises and automobile maintenance enterprises.*

| Notifications | Definitions |
|---------------|---|
| и | Utility of the consumer. |
| $u(\cdot)$ | Utility function. |
| Ct | Consumption of the consumer at time <i>t</i> . |
| K_t | Capital stock at time <i>t</i> . |
| K_{1t} | Capital used in automobile manufacturing production at time <i>t</i> . |
| K_{2t} | Capital used in automobile maintenance production at time <i>t</i> . |
| L | Fixed labor supply per period. |
| L_{1t} | Labor used in automobile manufacturing production at time <i>t</i> . |
| L_{2t} | Labor used in automobile maintenance production at time <i>t</i> . |
| β | Discount rate. |
| δ | Depreciation rate of capital. |
| p_t | Price of the automobile manufacturing production. |
| w_t | Wages of the labor. |
| r_t | Rent of capital. |
| T_t | Lump-sum tax on consumer's income. |
| λ_t | Lagrangian multiplier in consumer's problem of each period. |
| $F_1(\cdot)$ | Production functions of automobile manufacturing firm. |
| $F_2(\cdot)$ | Production functions of automobile maintenance firm. |
| π_1 | Profit of automobile manufacturing firm. |
| π_2 | Profit of automobile maintenance firm. |
| s_{1t} | Government subsidy ratios for automobile manufacturing firm. |
| s_{2t} | Government subsidy ratios for automobile maintenance firm. |
| а | Government's short-term GDP incentive intensity. |
| u_g | Utility of the government. |
| Γ | Lagrangian equation notation. |
| η_{1t} | Lagrange multipliers for automobile maintenance market constraint of each period. |
| η_{2t} | Lagrange multipliers for automobile manufacturing market constraint of each period. |
| η_{3t} | Lagrange multipliers for labor market constraint of each period. |
| ϕ | Lagrange multipliers for consumer's budget constraint. |
| Α | Constant, and $A = u'(c_0)[r_0K_0 + p_0(1 - \delta)K_0]$ |

Table 1. Notation list.

3. Empirical Method

3.1. Setting up of Empirical Model

According to the theoretical model, it can be known that the greater the government's short-term GDP incentives, the greater the difference in preferential subsidies between the two kinds of firms. Since most of the preferential subsidies are invisible, we use indirect methods to measure the biased intervention policies of the government. Specifically, we use the difference between the actual income tax rates in the two kinds of firms to reflect the difference in government intervention bias. The gap of actual income tax rate and the statutory tax rate, induced by taxation intensity, can reflect the degree of government intervention. The lower the actual income tax rate, the greater the government's preferential subsidy. Since the pre-tax profit is usually not the real profit of the enterprise, the proportion of the income tax in pre-tax profit does not accurately reflect the actual income tax rate of the enterprise. Some researchers use tax avoidance to measure the actual tax rate of the firms [41,42], because the larger the tax evasion equals the lower actual tax rate, which also indicates the smaller government

taxation intensity and the larger extent of the government intervention. We use the method of [41,42] to calculate the tax avoidance degree of the enterprise.

The idea of this method is like this. Although the reported profit is not the real profit of the enterprise, it is related to the real profit, and we can calculate the estimated profit, which is also not exactly the real profit, but related to it. Then there will be a large error if we use the difference between the estimated profit and the reported profit as the hidden profit of the enterprise. However, we can know that the estimated profit is related to the reported profit, and the greater the correlation, the smaller degree of the enterprise's tax avoidance. Thus, [41] measured the degree of tax avoidance of enterprises by calculating the sensitivity of estimated profit, which is based on the method of national income accounting, and the reported profit. Firstly, the estimated profit of the enterprise is highly correlated with the real profit π_{it} . It is assumed that the relationship between the real profit and the estimated profit is expressed as:

$$\pi_{it} = \eta_{it} + PRO_{it} + \theta_{it},\tag{29}$$

here, the subscripts *i* and *t* represent the enterprise and the year, respectively. η_{it} is a constant, θ_{it} is a random error, and *PRO* is the estimated profit. According to the principle of national income accounting, the estimated profit can be expressed as:

$$PRO_{it} = \frac{Y_{it} - MED_{it} - FC_{it} - WAGE_{it} - DEP_{it} - VAT_{it}}{TAS_{it}},$$
(30)

here, *Y* represents the total industrial output value of the enterprise; *MED* represents the total industrial input; *FC* represents the financial expenses; *WAGE* represents the expenditure of wage expressed by the total payable wage; *DEP* is the depreciation this year; *VAT* is the amount of payable value added tax and *TAS* indicates the total assets of the enterprise.

Secondly, the industrial enterprise database discloses the pre-tax profit reported by the company, which is the reported profit of the enterprise, denoted by *RPRO*, and the reported profit of the enterprise is also related to the real profit π_{it} , which can be expressed as:

$$RPRO_{it} = d_{it}\pi_{it} + e_{it} + \xi_{it}, \tag{31}$$

here the reported profit *RPRO* is defined as the ratio of total profit and total assets; e_{it} is the constant term; ξ_{it} is the random disturbance term; d_{it} is the sensitivity between the reported profit and the real profit, and the smaller value of d indicating the higher the degree of tax avoidance. According to the theoretical model, the government's GDP incentives will affect the difference between the actual tax burden of automobile manufacturing enterprises and automobile maintenance enterprises, which is the difference of sensitivity between the two kinds of firms. Therefore, we can express this impact as:

$$d_{it} = \beta_0 + \beta_1 V EHM_{it} + \beta_2 GOV_{it} + \beta_3 V EHM_{it} \times GOV_{it} + \chi X_{it} + \lambda_t + \gamma_i + \nu_{it},$$
(32)

here, *GOV* indicates the government's incentive; *VEHM* is a dummy variable which equals one if the firm belongs to the automobile manufacturing industry and zero otherwise; *X* indicates the firm level characteristic variables; λ_t represents the year fixed effect, and γ_j represents province fixed effect. β_1 presents the difference of sensitivity of reported profits to real profits between the manufacturing industry and the maintenance industry; β_2 indicates the influence of government incentives on the sensitivity of reported profits and real profits, and the coefficient β_3 of the interaction term *VEHM* × *GOV* indicates the difference of the impact of government's GDP incentives on sensitivity between automobile manufacturing firms and maintenance firms. Since the real profit of the enterprise cannot be observed, Equation (32) cannot be estimated directly. Therefore, we substitute the real profit represented by the estimated profit, which is Equation (29), into the expression of the reported profit, which is Equation (31) and then we can get

$$RPRO_{it} = d_{it}PRO_{it} + d_{it}\eta_{it} + e_{it} + \varepsilon_{it},$$
(33)

here, $\varepsilon_{it} = \xi_{it} + d_{it}\theta_{it}$. Then we take the expression of the sensitivity d_{it} between the profit and the real profit, which is Equation (32), into the relationship between the estimated profit and the reported profit, which is Equation (33) and we finally get this expression:

$$RPRO_{it} = (\beta_0 + \beta_1 VEHM_{it} + \beta_2 GOV_{it} + \beta_3 VEHM_{it} \times GOV_{it} + \beta_4 X_{it} + \lambda_t + \gamma_j) \times PRO_{it} + \alpha_0 + \alpha_1 VEHM_{it} + \alpha_2 GOV_{it} + \alpha_3 VEHM_{it} \times GOV_{it} + \alpha_4 X_{it} + \lambda_t + \gamma_j + \mu_{it}.$$
(34)

Thus, the coefficient β_3 of the triple interaction term $VEHM \times GOV \times PRO$ indicates the difference of the impact of government's GDP incentives on sensitivity between automobile manufacturing firms and automobile maintenance firms. If β_3 is negative, it means that automobile manufacturing firms have been given more preferential support under government incentives, which lead to their larger tax avoidance compared with the automobile maintenance companies.

Referring to the relevant literature [41,42], we control other firm level variables, including the firm size (*SIZE*), which is expressed in terms of the natural logarithm of the company's total assets; the liability-asset ratio (*LEV*), which equals the ratio of total liabilities to total assets of an enterprise; the age of the firm (*AGE*), which is the logarithm of the difference between the establishment year and the sample year; the financial expense (*LOAN*), which is the ratio of interest expense to total assets; the proportion of exports (*EXPT*), which is the ratio of export delivery value to the total assets; the ratio of industrial sales to the total value of industrial output (*SALE*) and the dummy variable (*SOE*) that indicates whether the enterprise is a state-owned enterprise or not.

3.2. Variable Definition of Government Incentive

Literature on political promotion tournaments, fiscal decentralization, fiscal revenue incentives and GDP growth [34,36] all mentioned that fiscal decentralization leads to GDP and fiscal incentives of the local government. In addition, stronger fiscal autonomy is most directly related to GDP incentives compared with other policies that can be used by local governments. Since China introduced to reform and opening policies, economic growth has been the main task of the government. When the promotion of officials depends more on the performance of economic growth under the horizontal competition among regions, the GDP incentive of local governments through fiscal approach becomes stronger [34,35]. Therefore, fiscal decentralization is the best measure of GDP incentive of local governments. It can be considered that the fiscal decentralization is a measure of autonomy of local governments on fiscal revenue and expenditure, and the larger the degree of fiscal decentralization, the stronger the fiscal autonomy of local governments because of the higher financial autonomy and the higher resource utilization capacity.

The fiscal decentralization system reflects the division of fiscal power and administrative power between the central and local governments, especially the revenue and expenditure power delegated by the central government to local ones. So the fiscal decentralization index should reflect the degree of local governments' control over fiscal power and administrative power. Oates firstly adopted three indicators of fiscal revenue and expenditure to represent the degree of fiscal decentralization [58]. Zhang and Gong [35] defined a group of indicators that represent the ratio of local government's income, or expenditure to the central government ability to measure fiscal decentralization. Lv et al. [42] used the tax decentralization index to measure the tax sharing incentives of local government. Since fiscal revenue, especially tax revenue, can represent a large part of the government's financial power, thus, tax sharing incentive is consistent with fiscal decentralization. So we use the following three tax sharing indicators to define the government's GDP incentive referring to Lv et al. [42].

- 1. The sharing rate of value added tax, denoted as *VAT*—which is defined as the ratio of the *VAT* revenue in the fiscal revenue of the province to the *VAT* revenue collected by the tax department of the province.
- 2. The sharing rate of income tax, denoted as *INC*—which is defined as the ratio of income tax revenue in the fiscal revenue of the province to the income tax revenue collected by the tax department of the province.
- 3. The sharing rate total tax, denoted as *TAX*—which is defined as the ratio of the total tax revenue in the fiscal revenue of the province to the total tax revenue collected by the tax department of the province.

3.3. Data

The data in our research comes from two resources. Firstly, the firm level data of automobile manufacturing and maintenance enterprises come from the database of Chinese Industrial Enterprises from 1998 to 2007. Researches using this database usually only cover these years because the sample in this period is believed to be more valid and reasonable. After excluding samples with missing values and samples with negative values in some important enterprise characteristic variables (such as total assets, total liabilities, total profit and industrial sales value), we get a total of 44,268 observations. Secondly, the data of macroeconomic variables, which includes the total tax revenue, VAT revenue, the income tax revenue in the fiscal income of each province, the total tax revenue, VAT revenue, the income tax revenue collected by the tax authorities of each province that are all used to calculate the provincial government incentives, come from Wind database, Chinese Tax Yearbook and Chinese Fiscal Yearbook. Wind database is a large financial database in China.

4. Empirical Evidence

4.1. Descriptive Statistics

Table 2 presents the descriptive statistics of variables in our regression. We can see that 90% of firms in our sample are in the automobile manufacturing industry, which is consistent with the situation that the automobile industry is less developed in China. The mean of the sharing rate of value added tax, income tax and total tax are 23%, 63%, and 50%, respectively. The average logarithm of firm age is 2, and the mean of leverage is 1%, with the standard deviation of 2% and a maximum value of 10%. The average proportion of the exports is 11%, with the standard error of 38%. Moreover, 25% of firms are SOEs and 6% foreign firms in our sample; 97% of firms are in normal operation; and 3% of the firms belong to the central government and 3% belong to large firms.

| Variables | Observations | Mean | SD | Min | Median | Max |
|-----------|--------------|------|------|-------|--------|-------|
| RPRO | 44,268 | 0.09 | 0.14 | 0.00 | 0.05 | 0.95 |
| PRO | 44,268 | 0.23 | 0.46 | -0.39 | 0.10 | 2.92 |
| VEHM | 44,268 | 0.90 | 0.30 | 0.00 | 1.00 | 1.00 |
| VAT | 44,268 | 0.23 | 0.03 | 0.19 | 0.23 | 0.27 |
| INC | 44,268 | 0.63 | 0.17 | 0.37 | 0.61 | 0.90 |
| TAX | 44,268 | 0.50 | 0.08 | 0.39 | 0.51 | 0.69 |
| SIZE | 44,268 | 9.87 | 1.65 | 0.69 | 9.67 | 18.17 |
| AGE | 44,268 | 2.00 | 0.99 | 0.00 | 1.95 | 7.60 |
| LOAN | 44,268 | 0.01 | 0.02 | 0.00 | 0.00 | 0.10 |
| EXPT | 44,268 | 0.11 | 0.38 | 0.00 | 0.00 | 2.40 |
| SALE | 44,268 | 0.97 | 0.08 | 0.67 | 0.99 | 1.27 |
| SOE | 44,268 | 0.25 | 0.44 | 0.00 | 0.00 | 1.00 |
| FOREG | 44,268 | 0.06 | 0.24 | 0.00 | 0.00 | 1.00 |
| OPRT | 44,268 | 0.97 | 0.18 | 0.00 | 1.00 | 1.00 |
| CENTR | 44,268 | 0.03 | 0.18 | 0.00 | 0.00 | 1.00 |
| LARGE | 44,268 | 0.03 | 0.18 | 0.00 | 0.00 | 1.00 |

Table 2. Descriptive statistics of each variable.

4.2. The Impact of Government Incentive on Tax Avoidance

According to the theoretical model, when the government's GDP incentive is large, the government will increase preferential treatment for enterprises, which will lower the actual tax burden of enterprises, and the government's preferential policies favors more to automobile manufacturing enterprises, which contribute more to capital accumulation. Table 3 shows the regression results of the impact of government incentives on the actual tax burden and the differences of this impact between automobile manufacturing and maintenance enterprises. We did not list the coefficients of intersection terms between the annual dummy variables and the calculated profit PRO in the table, and the intersection between provincial dummy variables and the calculated profit PRO are also omitted for simplicity. All regressions used the robust standard error to avoid heteroscedasticity problems. No government incentive variable is added In Column (1) of Table 2. The coefficient of intersection term of VEHM and *PRO* represents the difference of actual tax burden between automobile manufacturing enterprises and maintenance enterprises, and the coefficient is not significant, which means the actual tax burden difference between the two enterprises is not significant if the influence of government incentives is not controlled. The regression results of Column (2-4) show the regression results controlling the government incentives. The coefficients of the $GOV \times PRO$ in the column (2–4) are insignificant or negative, which indicates that the greater the government's GDP incentive is, the larger preferential to the enterprises, and thus, the actual tax burden of enterprises will be reduced. However, we care more about the coefficients of the triple intersection terms of $VEHM \times GOV \times PRO$. In the Column (2–4), the coefficients of these terms of are all significantly negative, which shows that although the government's GDP incentive leads to the reduction of the actual tax burden of enterprises, however, what more important is that the government's intervention is biased, and the government gives more preferential subsidies to firms of automobile manufacturing industries. The government's GDP incentive plays a bigger role in reducing the actual tax burden of enterprises for automobile manufacturing enterprises than for automobile maintenance enterprises. For example, the coefficient of the triple interaction term in Column (4) is -0.4934, which indicates that the negative effect of government incentives on firm's tax burden is about 0.5 lower in the automobile manufacturing firms compared with the automobile maintenance firms. These results confirm the hypothesis and prove the conclusion of theoretical model empirically as well.

| | Dependent Variable: RPRO | | | | | | |
|-----------|--------------------------|--------------------|--------------------|---------------|--|--|--|
| | (1) | (2) | (3) | (4) | | | |
| | No GOV | GOV = VAT | GOV = INC | GOV = TAX | | | |
| PRO_VEHM | -0.0166 | 0.1723 *** | 0.0578 * | 0.2273 *** | | | |
| | (-1.61) | (3.40) | (1.65) | (4.09) | | | |
| PRO_SIZE | 0.0011 | 0.0006 | 0.0002 | 0.0019 | | | |
| | (0.63) | (0.29) | (0.13) | (0.96) | | | |
| PRO_AGE | 0.0066 ** | 0.0072 ** | 0.0064 ** | 0.0064 ** | | | |
| | (2.11) | (2.32) | (2.05) | (2.07) | | | |
| PRO_LOAN | 0.1077 | 0.1132 | 0.1043 | 0.1257 | | | |
| | (1.19) | (1.25) | (1.16) | (1.39) | | | |
| PRO_EXPT | -0.0156 ** | -0.0155 ** | -0.0153 ** | -0.0167 *** | | | |
| | (-2.44) | (-2.39) | (-2.40) | (-2.61) | | | |
| PRO_SALE | 0.1015 *** | 0.0947 *** | 0.0763 *** | 0.1561 *** | | | |
| | (5.11) | (3.26) | (2.79) | (5.15) | | | |
| PRO_SOE | 0.0045 | 0.0036 | 0.0042 | 0.0050 | | | |
| | (0.61) | (0.49) | (0.57) | (0.68) | | | |
| PRO_FOREG | -0.0028 (-0.15) | -0.0023 (-0.12) | -0.0016 (-0.09) | 0.0029 (0.16) | | | |

Table 3. The influence of government incentive on the sensitivity of reported profit and real profit.

| | | Dependent Va | ariable: RPRO | |
|-------------------------|-------------|--------------|---------------|-------------|
| - | (1) | (2) | (3) | (4) |
| - | No GOV | GOV = VAT | GOV = INC | GOV = TAX |
| DDO | -0.0004 *** | -0.0005 *** | -0.0005 *** | -0.0006 *** |
| PRO | (-3.91) | (-4.56) | (-4.15) | (-5.16) |
| VELIM | 0.0111 *** | 0.0338 ** | 0.0015 | -0.0056 |
| VEHM | (5.66) | (2.23) | (0.16) | (-0.51) |
| CIZE | -0.0038 *** | -0.0037 *** | -0.0036 *** | -0.0039 *** |
| SIZE | (-9.52) | (-9.08) | (-8.97) | (-9.65) |
| 1.07 | -0.0090 *** | -0.0091 *** | -0.0089 *** | -0.0090 *** |
| AGE | (-14.65) | (-14.78) | (-14.60) | (-14.70) |
| LOUI | -0.0008 *** | -0.0006 *** | -0.0007 *** | -0.0007 *** |
| LOAN | (-3.34) | (-2.78) | (-3.16) | (-2.84) |
| | 0.0091 *** | 0.0091 *** | 0.0091 *** | 0.0094 *** |
| EXPT | (4.45) | (4.37) | (4.49) | (4.67) |
| 6 4 F F | 0.0015 | 0.0015 | 0.0016 | 0.0013 |
| SALE | (1.29) | (1.30) | (1.31) | (1.25) |
| 207 | -0.0091 *** | -0.0091 *** | -0.0090 *** | -0.0088 *** |
| SOE | (-5.98) | (-5.98) | (-5.89) | (-5.78) |
| FORES | 0.0169 *** | 0.0165 *** | 0.0167 *** | 0.0166 *** |
| FOREG | (5.40) | (5.25) | (5.31) | (5.30) |
| DRO VEUM COV | | -0.7748 *** | -0.0953 ** | -0.4750 *** |
| PRO_VEHM_GOV | | (-3.73) | (-2.19) | (-4.31) |
| VEUN CON | | -0.1003 | 0.0126 | 0.0323 |
| VEHM_GOV | | (-1.59) | (1.08) | (1.53) |
| | | -0.0671 | 0.0514 | -0.2954 *** |
| PRO_GOV | | (-0.45) | (1.40) | (-3.23) |
| 201/ | | 0.1660 ** | -0.0115 | -0.5370 *** |
| GOV | | (2.34) | (-0.95) | (-6.04) |
| <i>C</i> | 0.1450 *** | 0.1078 *** | 0.1507 *** | 0.3666 *** |
| Constant | (27.96) | (6.53) | (14.00) | (9.97) |
| YEAR | YES | YES | YES | YES |
| PRO * YEAR | YES | YES | YES | YES |
| PROVINCE | YES | YES | YES | YES |
| PRO * PROVINCE | YES | YES | YES | YES |
| Observations | 44,268 | 44,268 | 44,268 | 44,268 |
| Adjusted R ² | 0.35 | 0.35 | 0.35 | 0.35 |

Table 3. Cont.

Note: *t* statistics in parentheses, * p < 0.10, ** p < 0.05, *** p < 0.01.

4.3. The Influence of Government Incentive Changes on Tax Evasion of Different Types of Enterprises

4.3.1. Enterprises Ownership

The impact difference in government incentives varies in different types of enterprises. For example, it is generally believed that government interventions are different in enterprises with different ownership in the literature. On the one hand, state-owned enterprises will receive more preferential treatment and subsidies from the government. On the other hand, state-owned enterprises (SOEs) are mostly in the upstream industries, especially the manufacturing industry [59]. According to the theoretical model, enterprises in the automobile manufacturing industry, which is more likely to be the upstream industries, are more easily affected by the government incentives. So we conjecture that the difference of impact of government's GDP incentive tax avoidance between automobile manufacturing

enterprise and automobile maintenance enterprise will be more significant in state-owned enterprises. Therefore, we divided the samples into SOEs and non-SOEs to compare this impact difference. We define the SOEs as the firms whose registration types are state-owned, collective or wholly state-owned firms. Table 4 showed the regression results of Equation (28) in SOEs and non-SOEs. We can see that the coefficients of triple interaction term $VEHM \times GOV \times PRO$ are all significantly negative in the group of SOEs, which is Column (1) (3) and (5), and the absolute value is larger than the coefficients in Table 3. However, the coefficients of triple interaction term $VEHM \times GOVPRO$ are not significant in the group of non-SOEs, i.e., Column (2) (4) and (6), which indicates that the bias of preferential government subsidies caused by GDP incentives is more obvious in SOEs. The government, which is inspired by GDP incentive, gives more subsidizes to upstream enterprises, especially state-owned enterprises. This behavior will induce a larger amount of investment in industries that are more helpful to capital accumulation, which will promote economic growth. These results further prove our conclusion in the theoretical model.

Table 4. Regression results of different ownership enterprises.

| | Dependent Variable: RPRO | | | | | | |
|--------------|--------------------------|-------------|-------------|-------------|-------------|------------|--|
| | GOV | = VAT | GOV | | GOV | = TAX | |
| | (1) | (2) | (2) (3) (4) | | (5) (6) | | |
| | SOE | nSOE | SOE | nSOE | SOE | nSOE | |
| PRO_VEHM_GOV | -2.2074 *** | 0.5450 ** | -0.3651 *** | 0.2024 *** | -0.7655 *** | 0.0538 | |
| | (-6.28) | (1.98) | (-5.73) | (3.28) | (-4.78) | (0.33) | |
| VEHM_GOV | -0.0227 | -0.1985 ** | 0.0580 *** | -0.0606 *** | 0.0050 | 0.0157 | |
| | (-0.26) | (-2.09) | (3.77) | (-3.61) | (0.17) | (0.45) | |
| PRO_GOV | 0.1401 | -0.4679 ** | 0.1278 ** | -0.1619 *** | 0.1082 | -0.4560 ** | |
| | (0.50) | (-2.55) | (2.20) | (-3.08) | (0.66) | (-4.13) | |
| PRO_VEHM | 0.5187 *** | -0.1198 * | 0.2628 *** | -0.1511 *** | 0.3582 *** | -0.0102 | |
| | (5.98) | (-1.86) | (5.23) | (-3.05) | (4.51) | (-0.12) | |
| PRO_SIZE | -0.0016 | 0.0022 | -0.0022 | 0.0025 | -0.0010 | 0.0033 | |
| | (-0.42) | (0.95) | (-0.61) | (1.09) | (-0.26) | (1.42) | |
| PRO_AGE | -0.0017 | 0.0085 ** | -0.0011 | 0.0088 ** | -0.0024 | 0.0089 ** | |
| | (-0.30) | (2.26) | (-0.20) | (2.36) | (-0.42) | (2.39) | |
| PRO_LOAN | 0.4067 ** | -0.0237 | 0.3307 ** | -0.0175 | 0.4190 ** | -0.0094 | |
| | (2.42) | (-0.22) | (2.09) | (-0.16) | (2.52) | (-0.09) | |
| PRO_EXPT | -0.0308 | -0.0141 ** | -0.0129 | -0.0142 ** | -0.0254 | -0.0151 * | |
| | (-1.58) | (-2.11) | (-0.60) | (-2.12) | (-1.29) | (-2.25) | |
| PRO_SALE | 0.0169 | 0.1773 *** | 0.0169 | 0.1842 *** | 0.0140 | 0.2226 *** | |
| | (0.32) | (5.35) | (0.34) | (5.60) | (0.25) | (6.69) | |
| PRO | -0.0008 *** | 0.0002 | -0.0007 *** | 0.0003 | -0.0009 *** | 0.0005 | |
| | (-5.07) | (0.12) | (-4.69) | (0.14) | (-5.49) | (0.25) | |
| VEHM | 0.0159 | 0.0585 *** | -0.0323 *** | 0.0597 *** | 0.0092 | 0.0030 | |
| | (0.73) | (2.59) | (-2.68) | (4.55) | (0.63) | (0.16) | |
| GOV | 0.0413 | 0.1610 | -0.0498 *** | 0.0560 *** | -0.3299 ** | -0.4196 ** | |
| | (0.38) | (1.57) | (-3.15) | (3.11) | (-2.03) | (-3.87) | |
| SIZE | -0.0065 *** | -0.0024 *** | -0.0064 *** | -0.0024 *** | -0.0066 *** | -0.0026 ** | |
| | (-9.45) | (-4.74) | (-9.60) | (-4.83) | (-9.62) | (-5.17) | |
| AGE | -0.0113 *** | -0.0077 *** | -0.0113 *** | -0.0078 *** | -0.0112 *** | -0.0078 ** | |
| | (-11.01) | (-10.11) | (-11.02) | (-10.20) | (-10.93) | (-10.23) | |
| LOAN | 0.1052 ** | -0.0009 | 0.1197 *** | -0.0009 | 0.1277 *** | -0.0010 | |
| | (2.37) | (-0.73) | (2.77) | (-0.76) | (3.00) | (-0.82) | |
| EXPT | 0.0244 ** | 0.0082 *** | 0.0217 ** | 0.0082 *** | 0.0230 ** | 0.0084 *** | |
| | (2.30) | (3.76) | (2.05) | (3.69) | (2.11) | (3.83) | |
| SALE | -0.0000 | 0.0043 * | -0.0000 | 0.0042 * | -0.0000 | 0.0039 * | |
| | (-0.10) | (1.74) | (-0.09) | (1.74) | (-0.05) | (1.79) | |

| | Dependent Variable: RPRO | | | | | | |
|-------------------------|--------------------------|----------------------|-----------------------|----------------------|----------------------|----------------------|--|
| | GOV | = VAT | GOV | = INC | GOV = TAX | | |
| | (1) | (2) | (3) | (4) | (5) | (6) | |
| | SOE | nSOE | SOE | nSOE | SOE | nSOE | |
| PRO_FOREG | | -0.0054 (-0.29) | | -0.0070 (-0.37) | | -0.0044 (-0.23) | |
| FOREG | | 0.0163 *** (5.16) | | 0.0168 *** (5.28) | | 0.0164 *** (5.18) | |
| Constant | 0.1691 *** (6.88) | 0.0791 *** (3.29) | 0.2144 *** (14.37) | 0.0752 *** (5.06) | 0.3103 *** (4.63) | 0.2938 *** (6.46) | |
| YEAR | YES | YES | YES | YES | YES | YES | |
| PRO * YEAR | YES | YES | YES | YES | YES | YES | |
| PROVINCE | YES | YES | YES | YES | YES | YES | |
| PRO * PROVINCE | YES | YES | YES | YES | YES | YES | |
| Observations | 11,286 | 32,982 | 11,286 | 32,982 | 11,286 | 32,982 | |
| Adjusted R ² | 0.45 | 0.32 | 0.44 | 0.32 | 0.44 | 0.32 | |

Table 4. Cont.

Note: *t* statistics in parentheses, * *p* < 0.10, ** *p* < 0.05, *** *p* < 0.01.

4.3.2. Firm Size

Large enterprises playing a greater role in capital accumulation, and the government tend to "seize the large enterprises and release the small ones" [60], so large enterprises are more like to be the government's target and can promote GDP growth better. Therefore, if the government conducts a differential intervention on enterprises in different industries, due to the GDP incentive, we conjecture that the effect of a government tax incentive on reducing the actual tax burden of manufacturing enterprises will be more obvious in large enterprises. Therefore, we group the samples by firm size disclosed in the industrial enterprise database. The large firms include the enterprises with the label of "large scale", "super scale", "large type 1" and "large type 2", and the small firms include the enterprises with the label of "medium", "medium type 1", "medium type 2" and "small". Table 5 shows the regression results of the Equation (28) in firms with different size. We can see that all the coefficients of triple interaction terms are significantly negative, and the values are large in large firms' group. These results show that the phenomenon of the difference of intervention caused by government incentive is more apparent in large size enterprise, and the biased subsidies and preferential policies are given by the government to the automobile manufacturing industry are more likely to occur in large enterprises. These results prove our hypothesis, as well.

Table 5. Regression results of enterprises of different sizes.

| | Dependent Variable: RPRO | | | | | |
|--------------|--------------------------|-------------|-------------|-------------|-------------|-------------|
| | GOV | = VAT | GOV | = INC | GOV = TAX | |
| | (1) | (2) | (3) | (4) | (5) | (6) |
| | LARGE SMALL | | LARGE SMALL | | LARGE | SMALL |
| PRO_VEHM_GOV | -1.3817 *** | -1.1970 *** | -0.1259 *** | -0.0886 *** | -0.4356 *** | -0.3980 *** |
| | (-10.22) | (-4.54) | (-4.21) | (-3.17) | (-4.44) | (-7.65) |
| VEHM_GOV | 0.0956 * | -0.1238 | 0.0081 | 0.0160 ** | 0.0732 *** | 0.0270 *** |
| | (1.66) | (-1.58) | (0.88) | (2.20) | (3.28) | (2.59) |
| PRO_GOV | 0.3859 *** | 0.2338 | 0.2129 *** | 0.0745 *** | 0.0594 | 0.1101 ** |
| | (3.56) | (1.21) | (7.67) | (2.95) | (0.69) | (2.50) |
| PRO_VEHM | 0.3358 *** | 0.2780 *** | 0.0839 *** | 0.0531 ** | 0.2188 *** | 0.2030 *** |
| | (10.12) | (4.37) | (3.61) | (2.43) | (4.25) | (7.84) |

| | | | Dependent Va | ariable: RPRO | | |
|-------------------------|-------------|-------------|--------------|---------------|-------------|-----------|
| | GOV | = VAT | GOV | = INC | GOV | = TAX |
| | (1) | 1) (2) | (3) | (4) | (5) | (6) |
| | LARGE | SMALL | LARGE | SMALL | LARGE | SMALL |
| PRO_SIZE | -0.0139 *** | -0.0055 * | -0.0130 *** | -0.0036 ** | -0.0126 *** | -0.0076 * |
| | (-11.65) | (-1.75) | (-11.08) | (-2.21) | (-8.36) | (-7.17) |
| PRO_AGE | -0.0008 | 0.0068 | -0.0005 | 0.0062 *** | 0.0007 | 0.0016 |
| | (-0.48) | (1.63) | (-0.27) | (3.00) | (0.34) | (1.20) |
| PRO_LOAN | -0.1972 *** | 0.0786 | -0.1941 *** | 0.0961 | -0.1060 * | 0.1929 ** |
| | (-4.42) | (0.57) | (-4.28) | (1.17) | (-1.88) | (3.45) |
| PRO_EXPT | -0.0434 *** | 0.0074 | -0.0440 *** | 0.0001 | -0.0441 *** | 0.0032 |
| | (-11.67) | (0.63) | (-12.05) | (0.02) | (-9.74) | (0.77) |
| PRO_SALE | 0.2440 *** | 0.0438 | 0.2221 *** | 0.0522 *** | 0.2634 *** | 0.0746 ** |
| | (12.56) | (1.23) | (11.31) | (3.14) | (10.17) | (5.71) |
| PRO_SOE | -0.0218 *** | 0.0140 | -0.0267 *** | 0.0154 *** | -0.0222 *** | 0.0061 * |
| | (-5.24) | (1.49) | (-6.51) | (3.13) | (-4.38) | (1.82) |
| PRO_FOREG | 0.0620 *** | -0.0039 | 0.0538 *** | 0.0172 | 0.0418 *** | 0.0246 ** |
| | (6.01) | (-0.14) | (5.17) | (1.22) | (2.92) | (2.76) |
| PRO | -0.0043 | -0.0007 *** | -0.0022 | -0.0012 * | -0.0004 | -0.0017 * |
| | (-1.43) | (-5.23) | (-0.71) | (-1.76) | (-0.15) | (-5.04) |
| VEHM | -0.0204 | 0.0381 ** | -0.0036 | -0.0058 | -0.0333 *** | -0.0112 |
| | (-1.47) | (2.00) | (-0.49) | (-1.03) | (-2.88) | (-2.10) |
| GOV | -0.0523 | 0.2415 *** | -0.0217 ** | -0.0254 *** | -0.4713 *** | -0.0537 |
| | (-0.84) | (2.68) | (-2.30) | (-3.21) | (-6.59) | (-1.16) |
| SIZE | -0.0041 *** | -0.0118 *** | -0.0042 *** | -0.0076 *** | -0.0039 *** | -0.0065 * |
| | (-15.01) | (-15.75) | (-15.29) | (-19.22) | (-11.98) | (-25.58) |
| AGE | -0.0080 *** | -0.0079 *** | -0.0080 *** | -0.0062 *** | -0.0078 *** | -0.0042 * |
| | (-19.26) | (-9.42) | (-19.25) | (-12.19) | (-15.94) | (-12.46 |
| LOAN | 0.0009 | 0.1756 *** | -0.0005 | 0.1404 *** | -0.0016 | 0.1659 ** |
| | (0.45) | (3.80) | (-0.23) | (4.47) | (-0.98) | (9.09) |
| EXPT | 0.0195 *** | 0.0060 ** | 0.0194 *** | 0.0064 *** | 0.0169 *** | 0.0074 ** |
| | (13.61) | (2.36) | (13.60) | (4.96) | (7.77) | (8.42) |
| SALE | 0.0073 *** | 0.0004 | 0.0075 *** | 0.0002 | 0.0089 *** | 0.0025 * |
| | (3.06) | (0.54) | (3.01) | (0.40) | (2.87) | (1.82) |
| SOE | -0.0089 *** | -0.0036 * | -0.0088 *** | -0.0053 *** | -0.0100 *** | -0.0029 * |
| | (-8.87) | (-1.65) | (-8.75) | (-4.03) | (-8.40) | (-3.36) |
| FOREG | 0.0185 *** | 0.0214 *** | 0.0195 *** | 0.0137 *** | 0.0238 *** | 0.0127 ** |
| | (9.32) | (4.55) | (9.77) | (5.29) | (9.44) | (7.40) |
| Constant | 0.1509 *** | 0.1355 *** | 0.1543 *** | 0.1557 *** | 0.3339 *** | 0.1345 ** |
| | (10.04) | (6.35) | (18.46) | (20.84) | (11.27) | (7.11) |
| YEAR | YES | YES | YES | YES | YES | YES |
| PRO*YEAR | YES | YES | YES | YES | YES | YES |
| PROVINCE | YES | YES | YES | YES | YES | YES |
| PRO*PROVINCE | YES | YES | YES | YES | YES | YES |
| Observations | 18,283 | 22,135 | 18,271 | 21,413 | 20,269 | 19,038 |
| Adjusted R ² | 0.66 | 0.34 | 0.66 | 0.44 | 0.56 | 0.56 |

Table 5. Cont.

Note: *t* statistics in parentheses, * p < 0.10, ** p < 0.05, *** p < 0.01.

18 of 25

4.3.3. Firm Belongs

The short-term GDP incentive of the government mainly refers to the incentive of local governments. The government tends to implement policies more effectively for enterprises that are close to it considering asymmetric information; thus, the enterprises far away are more likely to be decentralized to lower level of local governments [61]. Therefore, the local governments tend to give more intervention to enterprises that are in their jurisdiction for GDP growth. So we grouped the samples into local and non-local enterprises groups according to the level of enterprise registration authorities disclosed in the industrial enterprise database. Local enterprise refers to the firms whose enterprise registration authority is not a state. Table 6 shows the regression results in the different groups. We can see that the coefficient of triple interaction terms is all significantly negative in the local group, which are the Column (2) (4) and (6). However, they are insignificant in the group of central government authorized firms, which are the Column (1) (3) and (5). The results also suggest that the phenomenon of differential intervention caused by government incentives is more obvious in local firms. This may because the biased subsidies and preferential policies that the government gives to the automobile industry are more likely to occur in the local firm under jurisdiction.

| | | Dependent Variable: RPRO | | | | | | |
|--------------|-------------|--------------------------|-------------|-------------|-------------|-------------|--|--|
| | GOV | = VAT | GOV | = INC | GOV | = TAX | | |
| | (1) | (2) | (3) | (4) | (5) | (6) | | |
| | CENTR | LOCAL | CENTR | LOCAL | CENTR | LOCAL | | |
| PRO_VEHM_GOV | -0.2966 | -0.7750 *** | 0.0913 | -0.0811 *** | -0.0114 | -0.4638 *** | | |
| | (-0.48) | (-3.67) | (0.50) | (-2.70) | (-0.02) | (-4.06) | | |
| VEHM_GOV | 0.0353 | -0.0920 | 0.0066 | 0.0006 | 0.0325 | 0.0371 | | |
| | (0.37) | (-1.37) | (0.28) | (0.08) | (0.53) | (1.60) | | |
| PRO_GOV | 0.1970 | -0.0716 | 0.1946 | 0.0566 ** | 0.5439 | -0.2408 *** | | |
| | (0.42) | (-0.47) | (1.19) | (2.25) | (1.24) | (-2.60) | | |
| PRO_VEHM | 0.0500 | 0.1627 *** | -0.1222 | 0.0339 | -0.0438 | 0.2126 *** | | |
| | (0.35) | (3.15) | (-0.93) | (1.42) | (-0.20) | (3.68) | | |
| PRO_SIZE | 0.0141 *** | 0.0004 | 0.0191 ** | 0.0015 | 0.0177 ** | 0.0013 | | |
| | (4.15) | (0.17) | (2.24) | (1.18) | (1.99) | (0.64) | | |
| PRO_AGE | 0.0146 * | 0.0070 ** | -0.0038 | 0.0099 *** | -0.0043 | 0.0063 ** | | |
| | (1.86) | (2.21) | (-0.25) | (5.17) | (-0.27) | (1.98) | | |
| PRO_LOAN | -1.7017 ** | 0.0874 | -1.8410 | 0.1293 * | -1.8837 | 0.1005 | | |
| | (-2.16) | (0.97) | (-0.82) | (1.89) | (-0.88) | (1.11) | | |
| PRO_EXPT | -0.0951 | -0.0161 ** | -0.0588 | -0.0184 *** | -0.0492 | -0.0171 *** | | |
| | (-1.39) | (-2.50) | (-0.59) | (-4.16) | (-0.48) | (-2.69) | | |
| PRO_SALE | -0.0410 | 0.1053 *** | -0.1465 | 0.0577 *** | -0.1975 | 0.1510 *** | | |
| | (-0.41) | (3.56) | (-1.12) | (3.26) | (-1.22) | (4.92) | | |
| PRO_SOE | -0.1054 *** | 0.0069 | -0.1440 *** | -0.0001 | -0.1482 *** | 0.0073 | | |
| | (-3.98) | (0.93) | (-3.37) | (-0.03) | (-3.38) | (0.99) | | |
| PRO | -0.0318 *** | -0.0005 *** | -0.0204 | -0.0003 *** | -0.0208 | -0.0006 *** | | |
| | (-3.05) | (-4.72) | (-0.93) | (-6.30) | (-0.92) | (-5.35) | | |
| VEHM | -0.0029 | 0.0327 ** | 0.0015 | 0.0086 | -0.0102 | -0.0071 | | |
| | (-0.12) | (2.02) | (0.08) | (1.42) | (-0.35) | (-0.59) | | |
| GOV | -0.1764 | 0.1571 ** | -0.0345 | -0.0037 | 0.1539 | -0.5295 *** | | |
| | (-1.28) | (2.09) | (-1.31) | (-0.46) | (0.77) | (-5.80) | | |
| SIZE | -0.0011 * | -0.0039 *** | -0.0025 * | -0.0010 *** | -0.0024 | -0.0041 *** | | |
| | (-1.68) | (-8.86) | (-1.72) | (-3.34) | (-1.56) | (-9.24) | | |
| AGE | -0.0041 *** | -0.0091 *** | -0.0061 *** | -0.0085 *** | -0.0063 *** | -0.0091 *** | | |
| | (-3.22) | (-14.43) | (-2.71) | (-18.85) | (-2.79) | (-14.39) | | |

Table 6. Regression results of central and local enterprises.

| | Dependent Variable: RPRO | | | | | | | |
|-------------------------|--------------------------|------------------------|------------------------|------------------------|------------------------|------------------------|--|--|
| | GOV | = VAT | GOV | GOV = INC | | GOV = TAX | | |
| | (1) | (2) | (3) | (4) | (5) | (6) | | |
| | CENTR | LOCAL | CENTR | LOCAL | CENTR | LOCAL | | |
| LOAN | -0.0668 (-0.90) | -0.0006 *** (-2.61) | 0.1081 (0.43) | 0.0565 *** (2.63) | 0.0977 (0.40) | -0.0006 *** (-2.72) | | |
| EXPT | -0.0152 (-0.79) | 0.0093 *** (4.49) | -0.0381 (-1.57) | 0.0091 *** (5.89) | -0.0390 (-1.60) | 0.0096 *** (4.76) | | |
| SALE | 0.0009 (0.42) | 0.0015 (1.27) | 0.0039 (0.76) | 0.0016 (1.35) | 0.0040 (0.74) | 0.0013 (1.24) | | |
| SOE | -0.0220 *** (-6.03) | -0.0087 *** (-5.51) | -0.0158 *** (-2.94) | -0.0084 *** (-7.41) | -0.0156 *** (-2.83) | -0.0084 ** (-5.28) | | |
| PRO_FOREG | | -0.0012 (-0.07) | | 0.0228 ** (2.13) | | 0.0036 (0.19) | | |
| FOREG | | 0.0168 *** (5.29) | | 0.0147 *** (6.77) | | 0.0168 *** (5.34) | | |
| Constant | 0.1089 *** (3.70) | 0.1130 *** (6.40) | 0.1179 *** (4.95) | 0.1107 *** (15.34) | 0.0299 (0.37) | 0.3658 *** (9.68) | | |
| YEAR | YES | YES | YES | YES | YES | YES | | |
| PRO*YEAR | YES | YES | YES | YES | YES | YES | | |
| PROVINCE | YES | YES | YES | YES | YES | YES | | |
| PRO*PROVINCE | YES | YES | YES | YES | YES | YES | | |
| Observations | 1488 | 42,744 | 1524 | 41,775 | 1524 | 42,744 | | |
| Adjusted R ² | 0.30 | 0.35 | 0.21 | 0.43 | 0.21 | 0.35 | | |

Table 6. Cont.

5. Robustness Check

We do robustness check to make sure that our results are invalid. We used only the sample firms whose business status is normal operation. The business status of enterprises can be divided into business, closure, preparation, the closed, the bankruptcy, and other six states, according to the database of China's industrial enterprises. The enterprise in the state of abnormal operating may have biased tax avoidance, and the only enterprise in the state of normal operating can reflect the actual level of government intervention. Therefore, we only keep the firms who are in normal business status. There are 1492 companies are dropped, and Table 7 shows the regression results. We can see that the regression results are similar to the results in Table 3. For example, the first coefficient of triple interaction term $PRO \times VEHM \times VAT$ in Column (2) is -0.8267, which is significant. These results show that the intervention on automobile manufacturing firms is biased as well, which is consistent with our previous results.

| | (1) | (2) | (3) | (4) |
|----------|----------|------------|----------|------------|
| | RPRO | RPRO | RPRO | RPRO |
| PRO_VEHM | -0.0163 | 0.1854 *** | 0.0537 | 0.1997 *** |
| | (-1.57) | (3.57) | (1.53) | (3.43) |
| PRO_SIZE | -0.0003 | -0.0011 | -0.0013 | 0.0003 |
| | (-0.18) | (-0.52) | (-0.67) | (0.16) |
| PRO_AGE | 0.0059 * | 0.0066 ** | 0.0057 * | 0.0057 * |
| | (1.84) | (2.07) | (1.78) | (1.77) |

Table 7. Robust Check: Only firms in operation.

Note: *t* statistics in parentheses, * p < 0.10, ** p < 0.05, *** p < 0.01.

| - | (1) | (2) | (3) | (4) |
|--------------|-----------------------|------------------------|-----------------------|----------------------|
| | RPRO | RPRO | RPRO | RPRO |
| PRO_LOAN | -0.0698 | -0.0577 | -0.0720 | -0.0590 |
| | (-0.70) | (-0.58) | (-0.72) | (-0.58) |
| PRO_EXPT | -0.0250 *** | -0.0249 *** | -0.0246 *** | -0.0242 *** |
| | (-3.91) | (-3.85) | (-3.84) | (-3.76) |
| PRO_SALE | 0.1188 *** (5.86) | 0.1108 *** (3.80) | 0.0921 *** (3.40) | 0.1582 *** (5.24) |
| | 0.0066 | 0.0052 | 0.0060 | 0.0071 |
| PRO_SOE | (0.88) | (0.70) | (0.81) | (0.96) |
| PRO_FOREG | 0.0024 | 0.0031 | 0.0035 | 0.0050 |
| FRO_FORLG | (0.13) | (0.17) | (0.19) | (0.27) |
| PRO | -0.0003 | -0.0005 | -0.0006 | 0.0001 |
| | (-0.26) | (-0.38) | (-0.43) | (0.07) |
| VEHM | 0.0103 *** (5.16) | 0.0353 ** (2.30) | 0.0004 (0.05) | -0.0019 (-0.18) |
| | -0.0034 *** | -0.0033 *** | -0.0032 *** | -0.0035 *** |
| SIZE | (-8.39) | (-7.93) | (-7.81) | (-8.51) |
| 4.05 | -0.0090 *** | -0.0091 *** | -0.0090 *** | -0.0090 *** |
| AGE | (-14.57) | (-14.69) | (-14.55) | (-14.58) |
| LOAN | 0.1037 *** | 0.0989 *** | 0.1028 *** | 0.1034 *** |
| EO/IIV | (3.49) | (3.30) | (3.47) | (3.49) |
| EXPT | 0.0113 *** (5.63) | 0.0114 *** (5.53) | 0.0113 *** (5.65) | 0.0112 *** (5.55) |
| | 0.0013 | 0.0014 | 0.0015 | 0.0012 |
| SALE | (1.25) | (1.26) | (1.28) | (1.21) |
| | -0.0071 *** | -0.0070 *** | -0.0070 *** | -0.0072 *** |
| SOE | (-4.63) | (-4.59) | (-4.54) | (-4.69) |
| FOREG | 0.0162 *** | 0.0157 *** | 0.0159 *** | 0.0159 *** |
| TOREG | (5.19) | (5.01) | (5.08) | (5.10) |
| PRO_VEHM_GOV | | -0.8267 *** (-3.90) | -0.0897 ** (-2.05) | -0.4221 *** |
| | | -0.1098 * | 0.0128 | (-3.67) |
| VEHM_GOV | | (-1.73) | (1.11) | 0.0237 (1.13) |
| | | -0.0618 | 0.0543 | -0.2367 ** |
| PRO_GOV | | (-0.41) | (1.48) | (-2.51) |
| GOV | | 0.1651 ** | -0.0054 | -0.1055 |
| 007 | | (2.31) | (-0.45) | (-1.16) |
| Constant | 0.1398 *** (27.33) | 0.1024 *** (6.15) | 0.1408 *** (13.43) | 0.1873 *** (5.03) |
| | · , | (8.13) YES | . , | |
| YEAR | YES | | YES | YES |
| PRO*YEAR | YES | YES | YES | YES |
| PROVINCE | YES | YES | YES | YES |
| PRO*PROVINCE | YES | YES | YES | YES |
| Observations | 42,776 | 42,776 | 42,776 | 42,776 |

Table 7. Cont.

Note: *t* statistics in parentheses, * *p* < 0.10, ** *p* < 0.05, *** *p* < 0.01.

6. Conclusions

The automobile manufacturing industry developed rapidly in China; however, the automobile maintenance industry less developed. In addition to the factors from the demand and supply side in

the market, we tried to explain this fact from the perspective of government incentive and intervention in China. Firstly, we established a two-sector model with government incentives and government intervention, and then we analyzed the game between the government and market, and the optimal subsidy policy under the government's GDP incentive is obtained. The results of the theoretical model show that the government gives more preferential policies to automobile manufacturing firms compare to firms in automobile maintenance industries under short term GDP incentive. That's why the development of the automobile manufacturing and maintenance industry is unbalanced. Secondly, we use three indicators to represent the government GDP incentive under fiscal decentralization and test the differential impact of GDP incentive on tax avoidance of the two kinds of firms empirically. The empirical results show that the GDP incentive of the government caused by fiscal decentralization induced more preferential treatment to automobile manufacturing enterprises, and thus, increase their tax evasion degree, which proves the mechanism of government incentive in our theoretical model.

Understanding the incentive and implementation of industrial policy can help us understand the evolution mechanism of China's automobile industrial policy and automobile industrial structure better. Well-developed automobile maintenance industrial can improve customer loyalty that can help the automobile manufacture company survive in this highly competitive industry; with the increase of new kinds of vehicles and customer's requirements, it is necessary for the automotive service industry to constantly improve its development mode and introduce new technologies; and the balanced growth of automobile manufacturing and maintenance industry is one of the driving forces for market sustainability of automobile industry and sustainable regional growth. Based on these facts, we propose that, in an initial phase of the development of the automotive industry, the automobile manufacturing industry, which is upstream industry, should be encouraged to develop more than automobile maintenance industry, which is downstream industry, because the latter would have no reason to exist if the upstream industries did not exist. However, China may pay more attention to rebalancing the weight of the two industries after passing this first phase as economic develops. In this new phase, some measures,, such as reducing the short term GDP incentive of local government and making the performance evaluation more diversified, will lead to better policy that promotes the transformation and upgrading of the whole automobile industrial structure, even the whole industrial structure optimization and economic growth.

Author Contributions: All authors were involved in preparing the manuscript. T.W. and L.M. contributed to the design of the research framework, L.M. contributed to the establishment of the theoretical model and the writing of the manuscript. Q.D. and T.W. conducted empirical analysis.

Funding: This research was funded by the National Natural Science Foundation of China, grant number [71703022 and 71804181], the Fundamental Research Funds for the Central Universities in UIBE, grant number [15QD21, CXTD7-04 and CXTD8-06], School of Banking and Finance, UIBE, and the National Center for Mathematics and Interdisciplinary Sciences, CAS.

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A

Proof of the result of Equation (18).

Proof. The consumer's budget constraint when t = 0 is:

$$c_0 - w_0 L + T_0 = p_0 \bigg[\frac{r_0}{p_0} K_0 + (1 - \delta) K_0 - K_1 \bigg].$$
(A1)

The consumer's budget constraint when t = 1 is:

$$c_1 - w_1 L + T_1 = p_1 \bigg[\frac{r_1}{p_1} K_1 + (1 - \delta) K_1 - K_2 \bigg].$$
 (A2)

The consumer's budget constraint when t = 2 is:

$$c_2 - w_2 L + T_2 = p_2 \bigg[\frac{r_2}{p_2} K_2 + (1 - \delta) K_2 - K_3 \bigg].$$
 (A3)

From the Equation (5), we can get

$$\frac{\beta u'(c_{t+1})}{u'(c_t)} = \frac{p_t}{p_{t+1}} \frac{1}{r_{t+1}/p_{t+1} + (1-\delta)}.$$
(A4)

Therefore, we can know that $\frac{\beta u'(c_1)}{u'(c_0)} = \frac{p_0}{p_1} \frac{1}{r_1/p_1 + (1-\delta)}$. Multiply the left hand of Equation (A2) by $\frac{\beta u'(c_1)}{u'(c_0)}$ and multiply the right hand of Equation (A2) by $\frac{p_0}{p_1} \frac{1}{r_1/p_1 + (1-\delta)}$, and then we can get:

$$\frac{\beta u'(c_1)}{u'(c_0)}(c_1 - w_1 L + T_1) = p_0 \left[K_1 - \frac{K_2}{\frac{r_1}{p_1} + (1 - \delta)} \right].$$
(A5)

Multiply the left hand of Equation (A3) by $\frac{\beta^2 u'(c_2)}{u'(c_0)}$ and multiply the right hand of Equation (A3) by $\frac{p_1}{p_2} \frac{1}{r_2/p_2 + (1-\delta)} \times \frac{p_0}{p_1} \frac{1}{r_1/p_1 + (1-\delta)}$, and then we can get:

$$\frac{\beta^2 u'(c_2)}{u'(c_0)}(c_2 - w_2 L + T_2) = p_0 \bigg[\frac{K_2}{r_1/p_1 + (1-\delta)} - \frac{K_3}{[r_2/p_2 + (1-\delta)][r_1/p_1 + (1-\delta)]} \bigg],$$
(A6)

and so on. We add up all the results from Equation (A1), i.e., $(A1) + (A5) + (A6) + \dots$, and we can get

$$\sum_{t=0}^{\infty} \beta^t \, \frac{u'(c_t)}{u'(c_0)} (c_t - w_t L + T_t) = [r_0 K_0 + p_0 (1 - \delta) K_0]. \tag{A7}$$

This is the same with Equation (18) in Section 2. \Box

References

- 1. Ma, L.; Wu, M.; Tian, X.; Zheng, G.; Du, Q.; Wu, T. China's provincial vehicle ownership forecast and analysis of the causes influencing the trend. *Sustainability* **2019**, *11*, 3928. [CrossRef]
- Wu, S.; Gang, L. An empirical study of after-sales service relationship in China's auto industry. In Proceedings of the 1st International Conference on Mechanical Engineering and Material Science, Shanghai, China, 28–30 December 2012; Atlantis Press: Paris, France, 2012.
- Brunnermeier, S.B.; Martin, S.A. Interoperability costs in the US automotive supply chain. *Supply Chain Manag.* 2002, 7, 71–82. [CrossRef]
- 4. Crandall, R. Vertical integration and the market for repair parts in the United States automobile industry. *J. Ind. Econ.* **1968**, *16*, 212–234. [CrossRef]
- 5. Fujimoto, T. The Japanese automobile parts supplier system: The triplet of effective inter-firm routines. *Int. J. Automot. Technol. Manag.* **2001**, *1*, 1–34. [CrossRef]
- Hebei Provincial Committee of the China Council for the Promotion of International Trade. China Automotive Aftermarket Blue Book 2013–2017. 2013. Available online: http://www.199it.com/archives/384629.html (accessed on 28 June 2019).
- Research and Markets. China Automotive Aftermarket Report 2018–2025. Available online: https://www. prnewswire.com/news-releases/china-automotive-aftermarket-report-2018-2025-300772340.html (accessed on 30 June 2019).
- Brekelmans, M.; Chen, K.Y. China's Automotive Aftermarket: A Strategic Opportunity. *China Automot. Rev.* 2014, 2, 17–18.

- 9. Izogo, E.E.; Ogba, I.E. Service quality, customer satisfaction and loyalty in automobile repair services sector. *Int. J. Qual. Reliab. Manag.* **2015**, *32*, 250–269. [CrossRef]
- 10. Herrmann, A.; Xia, L.; Monroe, K.B.; Huber, F. The influence of price fairness on customer satisfaction: An empirical test in the context of automobile purchases. *J. Prod. Brand Manag.* **2007**, *16*, 49–58. [CrossRef]
- 11. Moosa, M.Y.; Hassan, Z. Customer perceived values associated with automobile and brand loyalty. *Int. J. Account. Bus. Manag.* **2015**, *3*, 99–115. [CrossRef]
- 12. Godlevskaja, O.; Iwaarden, J.V.; Wiele, T.V.D. Moving from product-based to service-based business strategies: Services categorisation schemes for the automotive industry. *Int. J. Qual. Reliab. Manag.* **2011**, *28*, 62–94. [CrossRef]
- 13. Subrahmanyam, K.; Reich, S.M.; Waechter, N.; Espinoza, G. Online and offline social networks: Use of social networking sites by emerging adults. *J. Appl. Dev. Psychol.* **2008**, *29*, 420–433. [CrossRef]
- 14. Shankar, V.; Smith, A.K.; Rangaswamy, A. Customer satisfaction and loyalty in online and offline environments. *Int. J. Res. Mark.* 2003, *20*, 153–175. [CrossRef]
- 15. Jorgenson, D.W.; Kevin, J.; Stiroh, K.J. U.S. Economic Growth at the Industry Level. *Am. Econ. Rev.* **2000**, *90*, 161–167. [CrossRef]
- Jiang, X.; Li, H. Service industry and China's economy: Correlation and potential of faster growth. *Econ. Res. J.* 2004, 1, 4–15.
- 17. Chen, S.Y.; Jefferson, G.H.; Zhang, J. Structural change, productivity growth and industrial transformation in China. *China Econ. Rev.* **2011**, *22*, 133–150. [CrossRef]
- 18. Peneder, M. *Structural Change and Aggregate Growth;* WIFO Working Paper; Austrian Institute of Economic Research: Vienna, Austria, 2002.
- 19. Syrquin, M.; Chenery, H. Three decades of industrialization. *World Bank Econ. Rev.* **1989**, *3*, 145–181. [CrossRef]
- 20. Duarte, M.; Restuccia, D. The role of structural transformation in aggregate productivity. *Q. J. Econ.* **2010**, 125, 129–173. [CrossRef]
- 21. International Road Federation. *IRF World Road Statistics* 2013; International Road Federation Press: Vernier, Switzerland, 2013.
- 22. Dargay, J.; Sommer, G.M. Vehicle ownership and income growth, worldwide: 1960–2030. *Energy J.* 2007, 28, 143–170. [CrossRef]
- 23. Kongsamut, P.; Rebelo, S.; Xie, D. Beyond balanced growth. Rev. Econ. Stud. 2001, 68, 869–882. [CrossRef]
- 24. Acemoglu, D.; Guerrieri, V. Capital deepening and non-balanced economic growth. *J. Political Econ.* **2008**, 116, 467–498. [CrossRef]
- 25. Ngai, L.R.; Pissarides, C.A. Structural change in a multisector model of growth. *Am. Econ. Rev.* **2007**, *97*, 429–443. [CrossRef]
- 26. Erokhin, V. Self-sufficiency versus security: How trade protectionism challenges the sustainability of the food supply in Russia. *Sustainability* **2017**, *9*, 1939. [CrossRef]
- 27. National Bureau of Statistics of China. *Statistical Yearbook of the World*; China Statistics Press: Beijing, China, 2018. Available online: http://www.stats.gov.cn/ztjc/ztsj/gjsj/ (accessed on 26 June 2019).
- 28. Weiss, J. Implementing industrial policy: How to choose? Oxf. Dev. Stud. 2018, 46, 71–82. [CrossRef]
- 29. Tommaso, M.R.D.; Tassinari, M.; Bonnini, S.; Marozzi, M. Industrial policy and manufacturing targeting in the US: New methodological tools for strategic policy-making. *Int. Rev. Appl. Econ.* **2017**, *31*, 681–703. [CrossRef]
- 30. Aghion, P.; Cai, J.; Mathias, D.; Du, L.; Harrison, A.; Legros, P. Industrial policy and competition. *Am. Econ. J. Macroecon.* **2015**, *7*, 1–32. [CrossRef]
- 31. Acemoglu, D.; Golosov, M.; Tsyvinski, A. Dynamic Mirrlees Taxation under Political Economy Constraints. *Rev. Econ. Stud.* **2010**, *77*, 841–881. [CrossRef]
- 32. Acemoglu, D.; Golosov, M.; Tsyvinski, A. Political economy of Ramsey taxation. *J. Public Econ.* **2011**, *95*, 467–475. [CrossRef]
- 33. Maskin, E.; Qian, Y.Y.; Xu, C.G. Incentives, scale economies and organization forms. *Rev. Econ. Stud.* 2000, 67, 359–378. [CrossRef]
- 34. Zhang, J. Capital formation, industrialization and economic growth: Understanding China's economic reform. *Econ. Res. J.* **2002**, *6*, 3–13.

- 35. Zhang, Y.; Gong, L. The Fenshuizhi reform, fiscal decentralization and economic growth in China. *China Econ. Q.* **2005**, *5*, 75–108.
- 36. Li, H.; Zhou, L.A. Political turnover and economic performance: The incentive role of personnel control in China. *J. Public Econ.* **2005**, *89*, 1743–1762. [CrossRef]
- 37. Lu, G.; Wang, Z.; Zhang, C. Research on the performance of subsidizing innovation for Chinese strategic emerging industry. *Econ. Res. J.* **2014**, *49*, 44–55.
- Zhang, J.; Zheng, W. How does government subsidy affect the binary margin of Chinese enterprises' export? World Econ. 2015, 38, 22–48.
- Yang, Y.; Wei, J.; Luo, L. Who is using government subsidies to innovate?-The combined regulating effect of ownership and distorted factor markets. *Manag. World* 2015, 75–86.
- 40. Zhu, J.; Han, F.; Lu, Z. Industrial policy, bank connections, and debt financing: An empirical research based on A-share listed companies. *Financ. Res. J.* **2015**, 176–191.
- 41. Cai, H.; Liu, Q. Competition and corporate tax avoidance: Evidence from Chinese industrial firms. *Econ. J.* **2009**, *119*, 764–795. [CrossRef]
- 42. Lv, B.; Ma, G.; Mao, J. Tax sharing and tax rate: From government to entrepreneurs. *Econ. Res. J.* **2016**, *51*, 13–28.
- 43. Hsieh, C.T.; Klenow, P.J. Misallocation and manufacturing TFP in China and India. *Q. J. Econ.* **2009**, 124, 1403–1448. [CrossRef]
- 44. Andrei, J.; Darvasi, D. Perspectives and challenges in financing the new common agricultural policy, a new paradigm. *J. Food Agric. Environ.* **2012**, *10*, 904–907.
- 45. Lucas, R.E. Trade and the diffusion of the industrial revolution. *Am. Econ. J. Macroecon.* **2009**, *1*, 1–25. [CrossRef]
- 46. Gao, T.; Ivolga, A.; Erokhin, V. Sustainable rural development in northern China: Caught in a vice between poverty, urban attractions, and migration. *Sustainability* **2018**, *10*, 1467.
- 47. Wu, T.; Zhang, M.; Ou, X. Analysis of future vehicle energy demand in China based on a gompertz function method and computable general equilibrium model. *Energies* **2014**, *7*, 7454–7482. [CrossRef]
- Wu, T.; Zhao, H.; Ou, X. Vehicle ownership analysis based on GDP per capita in China: 1963–2050. Sustainability 2014, 6, 4877–4899. [CrossRef]
- 49. Ou, X.; Zhang, X.; Chang, S. Scenario analysis on alternative fuel/vehicle for China's future road transport: Life-cycle energy demand and GHG emissions. *Energy Policy* **2010**, *38*, 3943–3956. [CrossRef]
- 50. Ou, X.; Zhang, X.; Chang, S. Alternative fuel buses currently in use in China: Life-cycle fossil energy use, GHG emissions and policy recommendations. *Energy Policy* **2010**, *38*, 406–418. [CrossRef]
- 51. Ou, X.; Yan, X.; Zhang, X. Life-cycle energy consumption and greenhouse gas emissions for electricity generation and supply in China. *Appl. Energy* **2011**, *88*, 289–297. [CrossRef]
- 52. Lin, C.; Wu, T.; Ou, X.; Zhang, Q.; Zhang, X.; Zhang, X. Life-cycle private costs of hybrid electric vehicles in the current Chinese market. *Energy Policy* **2013**, *55*, 501–510. [CrossRef]
- 53. Wu, T.; Ma, L.; Mao, Z.; Ou, X. Setting up charging electric stations within residential communities in current China: Gaming of government agencies and property management companies. *Energy Policy* **2015**, 77, 216–226. [CrossRef]
- 54. Chen, X.; Wu, T.; Zheng, R.; Guo, X. How vehicle market is segmented and influenced by subsidy policy: A theoretical study. *Transp. Res. Part A Policy Pract.* **2018**, *118*, 776–782. [CrossRef]
- 55. Ma, L.; Zhai, Y.; Wu, T. Operating Charging Infrastructure in China to Achieve Sustainable Transportation: The Choice between Company-Owned and Franchised Structures. *Sustainability* **2019**, *11*, 1549. [CrossRef]
- 56. Wu, T.; Shen, Q.; Xu, M.; Peng, T.; Ou, X. Development and application of an energy use and CO2 emissions reduction evaluation model for China's online car hailing services. *Energy* **2018**, *154*, 298–307. [CrossRef]
- 57. Wu, T.; Zhang, M.; Tian, X.; Wang, S.; Hua, G. Spatial differentiation and network externality in pricing mechanism of online car hailing platform. *Int. J. Prod. Econ.* **2019**, *219*, 275–283. [CrossRef]
- 58. Oates, W.E. Searching for Leviathan: An empirical analysis. Am. Econ. Rev. 1985, 75, 748–757.
- 59. Li, X.; Liu, X.; Wang, Y. *A Model of China of China's State Capitalism*; HKUST IEMS Working Paper Series 2015-12; HKUST Institute for Emerging Market Studies: Hong Kong, China, 2015.

- 60. Hsieh, C.T.; Song, Z. Grasp the Large, Let Go of the Small: The Transformation of the State Sector in China. *Brook. Pap. Econ. Act.* **2016**, 2015, 295–366.
- 61. Huang, Z.; Li, L.; Ma, G.; Xu, L.C. Hayek, local information, and commanding heights: Decentralizing state-owned enterprises in China. *Am. Econ. Rev.* **2017**, *107*, 2455–2478. [CrossRef]



© 2019 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).